Author's response to reviews

Title: Effects of continuous renal replacement therapy on renal inflammatory cytokines during extracorporeal membrane oxygenation in a porcine model.

Authors:

Yimin Hu (guuyeym@163.com)
Wenkui Yu (yudrnj@163.com)
Jialiang Shi (jlsfe@163.com)
Qiyi Chen (giyichen2008@yahoo.cn)
Juanhong Shen (medcc.sjh@gmail.com)
Zhiliang Lin (linzhiliang@163.com)
Changsheng He (hcs@sina.com)
Ning Li (liningrigs@vip.sina.com)
Jieshou Li (nyhzzx@public1.ptt.js.cn)

Version: 3 Date: 21 March 2013

Author's response to reviews: see over
Response to the reviewers

To reviewer 1:

1. the article has been revised linguistically by an English speaking people.

To reviewer 2

1. We have explained the procedure carried out in sham group and the animals were ventilated after tracheotomy and an internal diameter 6.0 cm tracheal tube placement.
2. the article has been revised in language especially in the introduction and discussion.

Response to the editor

1. We revised the article with colored lines.
2. We provided a point by point response to the reviewers.
3. We added a background section in the text.
4. We added a conclusion section at the end of text.
5. We added an acknowledgement and authors contributions in the text.
6. We added the competing interest in the text.